Author’s response to reviews

Title: In vitro antiproliferative and apoptotic activities of alpha-, gamma- and delta-tocotrienol isomers on human cancer cells

Authors:

Su-Wen Lim (lsw_424@hotmail.com)
Hwei-San Loh (Sandy.Loh@nottingham.edu.my)
Kang-Nee Ting (Kang-Nee.Ting@nottingham.edu.my)
Tracey D Bradshaw (Tracey.Bradshaw@nottingham.ac.uk)
Nazariah A Zeenathul (zeenathul@upm.edu.my)

Version: 2 Date: 24 September 2014

Author’s response to reviews:

Dear Chief Editor,

I have included the line and page numbers in the manuscript as requested. Many thanks.

Yours sincerely,
HS Loh